Author’s response to reviews

Title: An assessment of fishing communities around Lake Victoria, Uganda, as potential populations for future HIV vaccine efficacy studies.

Authors:

Noah Kiwanuka (nkiwanuka@gmail.com)
Juliet Mpendo (jmpendo@iavi.or.ug)
Annet Nalutaaya (analutaaya@iavi.or.ug)
Matthias Wambuzi (mwambuzi@iavi.or.ug)
Annet Nanvubya (ananyubya@iavi.or.ug)
Paul K Kitandwe (pkato@iavi.or.ug)
Enoch Muyanja (EMuyanja@iavi.or.ug)
Julius Ssempiira (jssempiriira@iavi.or.ug)
Apolo Balyegisawa (ABalyegisawa@iavi.org)
Ali Ssetaala (assetala@iavi.or.ug)

Version: 2
Date: 10 July 2014

Author’s response to reviews:

The Editor, July 10, 2014
BMC Public Health

Dear Sir/Madam

RE: SUBMISSION OF A REVISED VERSION OF MS: 1344447146135355

Please find enclosed a revised version of MS: 1344447146135355 article entitled “An assessment of fishing communities around Lake Victoria, Uganda, as potential populations for future HIV vaccine efficacy studies.”

The following changes have been made to the version submitted yesterday:
1. Line numberings have been added to the main text
2. Tables have been moved to the bottom of the manuscript after references.